Author’s response to reviews

Title: Air pollution dispersion from biomass stoves to neighboring homes in Mirpur, Dhaka, Bangladesh

Authors:

Anne Weaver (aweaver@buffalo.edu)
Emily Gurley (emilysgurley@gmail.com)
Christina Crabtree-Ide (crc8@buffalo.edu)
Henrik Salje (henrik.salje@pasteur.fr)
Eun-Hye Yoo (eunhye@buffalo.edu)
Lina Mu (linamu@buffalo.edu)
Nasrin Akter (moonni007@gmail.com)
Pavani Ram (pkram@buffalo.edu)

Version: 3 Date: 25 Mar 2019

Author’s response to reviews:

1. Consent

-- In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

This has been edited to specify written consent.

2. Funding

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

This has been updated to indicate that, “The funder had no role in the study design, data collection, analysis or interpretation of data, or in manuscript writing.”
3. Authors Contributions

-- We notice that authors HS are missing from the authors' contributions section. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

It looks as though HT should be HS. Please update this if this was an error.

-- Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Yes, HT should have been HS. This was an error and was corrected. I have edited this section to indicate all authors have approved this manuscript.

4. Clean Manuscript

The manuscript submitted is clean of all comments, tracked changes, highlights, etc.